Author's response to reviews

Title: Coexpression of VEGF-C and cyclooxygenase-2 and its Association With Lymphangiogenesis in Human Breast Cancer

Authors:
- Xiao-Hua Zhang (oncology0607@yahoo.com.cn)
- Du-Ping Huang (hdping2006@yahoo.com.cn)
- Gui-Long Guo (oncology2004@163.com)
- Guo-Rong Chen (glchen2007@yahoo.com.cn)
- Hu-Xiang Zhang (hxzhang1234@yahoo.com.cn)
- Jian-Ming Li (jmliwz@yahoo.com.cn)

Version: 2 Date: 20 December 2006

Author's response to reviews: see over
Dear Editors:

On behalf of my co-authors, I am submitting the enclosed material “Coexpression of VEGF-C and Cox-2 and its Association With Lymphangiogenesis in Human Breast Cancer” for possible publication in your journal. This paper is new. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. We believe the paper may be of particular interest to the readers of your journal: VEGF-C and COX-2 are coexpressed and significantly associated with lymphangiogenesis and prognosis in breast cancer, suggesting COX-2 may up-regulate VEGF-C expression and thus promote lymph node metastasis via lymphangiogenesis pathway in human breast cancer.

Correspondence should be addressed to Xiao-hua Zhang at the following address, email address, fax number, and phone:

Xiao-Hua Zhang

Institute: Department of Oncology, The First Affiliated Hospital of Wenzhou Medical College

Adress: No.2, Fu Xue Road, 325000, WenZhou, China

e-mail address: oncology0607@yahoo.com.cn

Fax: 86-577-88069555

Tel: 86-577-88078236

Thanks very much for your attention to our paper.

Sincerely yours,

Xiao-hua Zhang